\capstartfalse 
\begin{deluxetable}{lrrrr} 
\tabletypesize{\footnotesize} 
\centering 
\tablewidth{240pt} 
\tablecaption{Stellar Properties \label{tab-star}} 
\tablecolumns{5}  
\tablehead{ 
 \colhead{Source} 
&\colhead{$d$ (pc)} 
&\colhead{SpT} 
&\colhead{$M_{\ast}$/$M_{\odot}$} 
&\colhead{Ref} 
} 
\startdata 
Sz 65 & 150 & K7.0 & 0.76 $\pm$ 0.18 & 2 \\
Sz 66 & 150 & M3.0 & 0.31 $\pm$ 0.04 & 1 \\
J15430131-3409153 & 150 & ... & ... & ... \\
J15430227-3444059 & 150 & ... & ... & ... \\
J15445789-3423392 & 150 & M5.0 & 0.12 $\pm$ 0.03 & 1 \\
J15450634-3417378 & 150 & ... & ... & ... \\
J15450887-3417333 & 150 & M5.5 & 0.14 $\pm$ 0.03 & 2 \\
Sz 68 & 150 & K2.0 & 2.13 $\pm$ 0.34 & 2 \\
Sz 69 & 150 & M4.5 & 0.19 $\pm$ 0.03 & 1 \\
Sz 71 & 150 & M1.5 & 0.42 $\pm$ 0.11 & 1
\enddata 
\tablenotetext{}{References: (1) \cite{2014A&A...561A...2A}, (2) Alcal\'{a} et al. (in prep), (3) \cite{2013MNRAS.429.1001A}, (4) \cite{2011MNRAS.418.1194M}, (5) \cite{2008ApJS..177..551M}, (6) Cleeves et al. (in prep), (7) \cite{2015A&A...578A..23B}, (8) \cite{2008hsf2.book..295C}. (This table is available in its entirety in machine-readible form.)} 
\end{deluxetable} 
\capstartfalse 
